

RESEARCH ARTICLE

10.1029/2023GH000869

Notice: Manuscript Authored by Battelle Memorial Institute Under Contract Number DE-AC05-76RL01830 with the US Department of Energy. The US Government retains and the publisher. by accepting this article for publication, acknowledges that the US Government retains a non-exclusive, paid-up, irrevocable, world-wide license to publish or reproduce the published form of this manuscript, or allow others to do so for US Government purposes. The Department of Energy will provide public access to these results of federally sponsored research in accordance with the DOE Public Access Plan: (http://energy.gov/ downloads/doe-public-access-plan)

** The submitted manuscript has been created by UChicago Argonne, LLC, Operator of Argonne National Laboratory ("Argonne"). Argonne, a U.S. Department of Energy Office of Science laboratory, is operated under Contract No. DE-AC02-06CH11357. The U.S. Government retains for itself, and others acting on its behalf, a paid-up nonexclusive, irrevocable worldwide license in said article to reproduce, prepare derivative works, distribute copies to the public, and perform publicly and display publicly, by or on behalf of the Government. The Department of Energy will provide public access to these results of federally sponsored research in accordance with the DOE Public Access Plan. http://energy. gov/downloads/doe-public-access-plan

Key Points:

- Lake Michigan reduces daytime heat stress in Chicago, while urbanization primarily increases nighttime heat stress
- Urban and lake impacts on skin temperature are stronger than the corresponding impacts on air and wet bulb globe temperature
- Disparities in skin temperature across Chicago's community areas are higher

© 2023 Battelle Memorial Institute. UChicago Argonne, LLC and The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons

the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Urban Versus Lake Impacts on Heat Stress and Its Disparities in a Shoreline City

TC. Chakraborty¹, Jiali Wang², Yun Qian¹, William Pringle², Zhao Yang¹, and Pengfei Xue^{2,3}

¹Atmospheric, Climate, and Earth Sciences Division, Pacific Northwest National Laboratory, Richland, WA, USA, ²Environmental Science Division, Argonne National Laboratory, Lemont, IL, USA, ³Department of Civil, Environmental and Geospatial Engineering, Michigan Technological University, Houghton, MI, USA

Abstract Shoreline cities are influenced by both urban-scale processes and land-water interactions, with consequences on heat exposure and its disparities. Heat exposure studies over these cities have focused on air and skin temperature, even though moisture advection from water bodies can also modulate heat stress. Here, using an ensemble of model simulations covering Chicago, we find that Lake Michigan strongly reduces heat exposure (2.75°C reduction in maximum average air temperature in Chicago) and heat stress (maximum average wet bulb globe temperature reduced by 0.86°C) during the day, while urbanization enhances them at night (2.75 and 1.57°C increases in minimum average air and wet bulb globe temperature, respectively). We also demonstrate that urban and lake impacts on temperature (particularly skin temperature), including their extremes, and lake-to-land gradients, are stronger than the corresponding impacts on heat stress, partly due to humidity-related feedback. Likewise, environmental disparities across community areas in Chicago seen for skin temperature are much higher (1.29°C increase for maximum average values per \$10,000 higher median income per capita) than disparities in air temperature (0.50°C increase) and wet bulb globe temperature (0.23°C increase). The results call for consistent use of physiologically relevant heat exposure stress to accurately capture the public health implications of urbanization.

Plain Language Summary Cities along coastlines are affected by both urban processes and interactions between land and water. Previous studies on heat exposure in these cities primarily focused on temperature of the ground surface (or skin) and the air above that surface, generally overlooking the influence of moisture from water bodies on heat stress. In our study, using multiple coupled weather simulations, we show that during the day, Lake Michigan significantly reduces heat exposure and heat stress. However, at night, urbanization increases them. We also found that the effects of urbanization and the lake on temperature, especially skin temperature and its extreme, as well as the temperature difference between the lake and the land, are stronger than their impacts on heat stress. This is due to weaker coupling between daytime air and skin temperature, as well as to humidity-related factors. Moreover, we observed that environmental disparities in skin temperature are much higher than equivalent disparities in heat stress. These findings highlight the importance of using heat exposure metrics that are relevant to human physiology to accurately understand and inform the public health implications of urbanization.

1. Introduction

Due to increased urbanization adjacent to water bodies (Tibbetts, 2002), studies have frequently examined the complex interactions along the land-water interface and their impacts on urban weather and climate (Hu & Xue, 2016; Keeler & Kristovich, 2012; Sharma et al., 2018; Shepherd & Burian, 2003; Theeuwes et al., 2013). The thermostatic effect of water leads to a positive thermal gradient from water bodies to land during daytime since the land warms faster than the water, generating lake or sea breezes that flow from over the cooler water surface to the warmer and lower pressure land surface (Birch et al., 2015; S. T. K. Miller et al., 2003). The effect reverses at night with the land near water bodies being warmer (due to higher thermal inertia of water) than land in the interior and land breezes generated from the land toward the water (J. Yang et al., 2022). These interactions set up strong temperature gradients across the waterfront that can influence heat exposure for coastal and shore-line populations. Several studies have examined this coastal influence on urban heat exposure using air temperature or satellite-derived surface or skin temperature (Roth et al., 1989; Wu et al., 2019; J. Yang et al., 2022). Although skin and air temperature are coupled at large scales, pedestrians are exposed to ambient air temperature,



than disparities in air and wet bulb globe temperature

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

TC. Chakraborty, tc.chakraborty@pnnl.gov

Citation:

Chakraborty, TC., Wang, J., Qian, Y., Pringle, W., Yang, Z., & Xue, P. (2023). Urban versus lake impacts on heat stress and its disparities in a shoreline city. *GeoHealth*, 7, e2023GH000869. https:// doi.org/10.1029/2023GH000869

Received 26 MAY 2023 Accepted 1 NOV 2023

Author Contributions:

Conceptualization: TC. Chakraborty, Jiali Wang Data curation: TC. Chakraborty, Jiali Wang Formal analysis: TC. Chakraborty Investigation: TC. Chakraborty, Jiali Wang Methodology: TC. Chakraborty, Jiali Wang, Yun Oian, Zhao Yang Project Administration: Yun Qian Resources: Yun Oian Validation: TC. Chakraborty, Jiali Wang, William Pringle, Zhao Yang Visualization: TC. Chakraborty Writing - original draft: TC. Chakraborty Writing - review & editing: Jiali Wang, Yun Qian, William Pringle, Zhao Yang, Pengfei Xue

GeoHealth

not skin temperature, especially relevant within cities (Venter et al., 2021). Moreover, the physiological response to heat depends not only on air temperature, but also on relative humidity and wind speed (among other factors) (Anderson et al., 2013). For instance, all else remaining constant, more moisture in the air usually increases heat stress, which reduces the differential between human skin and core temperature; and thus the body's cooling efficiency (Im et al., 2017). Both urbanization and proximity to water bodies can modify ambient relative humidity and wind patterns. Urbanization can dry the air by modifying the city's surface energy balance, often referred to as the urban moisture or dry island effect (Chakraborty et al., 2022; X. Huang et al., 2021; Z. Wang et al., 2021), and impede wind flow by increasing surface roughness (Qian et al., 2022). On the other hand, breezes originating from water bodies can bring in more moisture during the day (S. T. K. Miller et al., 2003). Although previous urban modeling studies have considered physiologically relevant heat stress metric (Chakraborty, Newman, et al., 2023; K. Huang et al., 2021; Oleson et al., 2015; Zhao, 2021), they are generally not run with configurations or resolutions capable of resolving coastal gradients. Thus, there is a need to understand how the interaction between urbanization and adjacent water bodies impact air temperature, relative humidity, and wind, and thus heat stress, in shoreline cities.

Due to the heterogeneity of cities, heat exposure shows large intra-urban variability. Across a global subset of cities (Chakraborty et al., 2019), and particularly in the US (Hsu et al., 2021), this variability leads to disproportionately higher heat exposure for poorer and other vulnerable communities. Some of these disparities relate to population dynamics within cities, frequently associated with historical discriminatory practices (Hoffman et al., 2020) as well as present inequities (Chakraborty, Newman, et al., 2023; Juday, 2015). Most multi-city studies that have examined disparities in heat exposure have done so using satellite-derived skin temperature (Benz & Burney, 2021; Chakraborty et al., 2019; Hoffman et al., 2020; Hsu et al., 2021). For land-locked cities, these disparities are strongly associated with inequities in urban vegetation, with richer populations living in greener areas (Chakraborty et al., 2020). However, cities along the waterfront are more complicated due to competing effects of neighborhood greenness and access to waterfronts on real estate prices (N. G. Miller et al., 2019), and thus on population distributions. Regarding the humidity contribution to heat stress, the moisture brought in by sea and lake breezes into neighborhoods can be a strong function of their proximity to the waterfront, while neighborhood-scale built-up properties (including urban vegetation) would also impact near-surface moisture content (Chakraborty et al., 2022; Qian et al., 2022). Although these complexities (the role of proximity to the water vs. neighborhood characteristics) are evident when examining satellite-derived skin temperature (Chakraborty et al., 2020), we do not know the bulk outcome of population distributions and meteorological variability within shoreline cities on potential disparities in heat stress.

Here, we conduct several sensitivity experiments using the Weather Research and Forecasting (WRF) model with multi-layer urban canopy representation and daily satellite-derived boundary conditions for lake surface temperature. These simulations are used to isolate the urban versus lake effect on the spatial variability of heat-related extremes within 77 community areas (neighborhoods henceforth) of Chicago during a typical summer (of 2018). We choose Chicago since it is a shoreline city (third largest among all US cities) to the west of Lake Michigan with a history of heat-related weather extremes (Conry et al., 2015; Kunkel et al., 1996). Previous studies have shown that the daytime breeze originating from Lake Michigan can penetrate 15-30 km inland, moving cooler and more humid air into Chicago during daytime, with an overall reduction in heat extremes (Conry et al., 2015; Keeler & Kristovich, 2012; Kunkel et al., 1996). However, these studies, and others that have investigated heat exposure in various shoreline cities (Roth et al., 1989; Wu et al., 2019; J. Yang et al., 2022), have mainly focused on skin or air temperature. Here, with the intent of focusing on physiologically relevant heat extremes, we also consider wet bulb globe temperature, the International Organization for Standardization (ISO) standard for occupational heat stress (Iso, 2017) that considers the combined impact of air temperature, relative humidity, wind speed, and solar radiation exposure on human health (Heo et al., 2019). The neighborhoods in Chicago show large residential segregation, with the richest neighborhoods lying along the waterfront (Figure 1a). Thus, we also examine disparities in heat stress and heat exposure across these neighborhoods and how they are impacted by the lake-to-land gradients. Our results show that although the lake and urbanization distinctly impact the magnitudes and gradients of air and skin temperature, these impacts flatten out for wet bulb globe temperature. The overall variability of maximum and minimum wet bulb globe temperature within Chicago is strongly controlled by air temperature, though wind speed and relative humidity also play detectable roles. Furthermore, heat exposure disparities and their extremes are generally more prominent when using skin temperature instead of air and wet bulb globe temperature, indicating the importance of using physiologically relevant variables when quantifying these environmental inequities.





Figure 1. Study area and overview schematic. (a) Spatial extent of Chicago neighborhoods (inset marks location of Chicago within the United States) and the median income per capita in each. (b) Conceptual schematic of Chicago in reference to Lake Michigan and rural Illinois with daytime and nighttime (corresponding to MODIS Aqua overpasses) lake-to-land gradients in skin temperature, air temperature, and wet bulb globe temperature derived from the five-member ensemble mean of WRF models simulations and the proximity of the neighborhood centroids to the lake shore. The numbers annotated with the colored gradients show the slopes of the best fit lines in °C per km distance from Lake Michigan. Note that the *x*-axis is reversed here (compared to Figure 4) for consistency with the schematic. The representation of Chicago is purely illustrative and not to scale.

2. Methods

2.1. Regions of Interest and Socioeconomic Data

Chicago is the third largest city in the United States, housing around 2.7 million people, and located on the southwestern side of Lake Michigan (Figure 1a). The city is affected by large pressure gradients during the summer due to the differential heating over the city and over Lake Michigan, resulting in strong lake breeze during the day (Laird et al., 2001). To examine spatial variability across this city, we consider the 77 community areas, which are commonly used to provide statistical summaries and for urban planning (Figure 1a). Each community area includes estimates of median income per capita (corresponding to U.S. census data for 2008–2012), which we extract from the City of Chicago's data portal (https://data.cityofchicago.org/). Since income is not a comprehensive proxy for vulnerable populations, we also consider the Hardship index, a more relevant metric of socioeconomic vulnerability that includes contributions from six factors—namely unemployment, lack of secondary education, per capita income, percentage population below poverty level, overcrowded housing, and age dependency—after normalization (Amdat, 2021). Overall, both Hardship index and median income per capita and lowest Hardship index (Figure 1a; Figure S4a in Supporting Information S1).

2.2. Model Simulations

We use Weather Research and Forecasting (WRF; version 4.3.1) model simulations (Skamarock & Klemp, 2008) centered at 45.5°N and 85.0°W and covering the Great Lakes region with a grid spacing of 4 km and 50 stretched vertical levels starting from 30.2 m above the ground (J. Wang et al., 2022) to simulate summertime conditions in Chicago. The initial and boundary conditions for the model (at 0.25° resolution every 3 hr) are taken from the European Centre for Medium-Range Weather Forecasts atmospheric reanalysis of the global climate, version 5 (ERA5) (Hersbach et al., 2020), while the boundary condition for the lake surface temperature (daily at 1.3 km) is from satellite-derived estimates from the National Oceanic and Atmospheric Administration (NOAA) Great Lakes Surface Environmental Analysis (Schwab et al., 1992). The model is run with the Rapid Radiative Transfer Model (RRTM) (Iacono et al., 2008) for shortwave and longwave radiation, the Yonsei University (YSU) planetary boundary layer scheme (Hong & Lim, 2006), revised Atmospheric Research Mesoscale Model Monin-Obukhov surface layer scheme (Jiménez et al., 2012), Thompson microphysics (Thompson et al., 2004), and the Unified Noah land surface model (Noah LSM) (F. Chen & Dudhia, 2001).

We run the model in three separate configurations: one corresponding to the present-day conditions (control), and two counterfactual scenarios with Lake Michigan (no lake) and Chicago urban grids (no urban) removed, respectively. For the control simulation, we use the latest multilayer urban canopy model (MLUCM) in WRF with Building Effect Parameterization (BEP) (Martilli et al., 2002) and Building Energy Model (BEM) (Salamanca et al., 2010). Together, BEP and BEM can represent the building impact on airflow and the exchange of energy between buildings and their surroundings. Grid-wise land cover is provided to the model from Moderate Resolution Imaging Spectroradiometer (MODIS) land cover with 21 classes, which WRF uses by default. The urban grids correspond to the high-density class of the MLUCM and have a fixed built-up fraction of 0.9. To examine the impact of urbanization and lake, respectively, the urban Chicago and Lake Michigan grids are replaced by cropland, the dominant land cover in the region. Thus, the difference between the control and no urban simulation provides the local urban effect, while the difference between the control and no lake simulation gives us the effect of Lake Michigan. This replacement of one surface type with another to examine the impacts on climate is a common approach in numerical modeling (K. Huang et al., 2021; Sarangi et al., 2021; Theeuwes et al., 2013). In contrast, for observational studies, such as those on urban heat islands, where a similar replacement of surface type is not methodologically possible, space-for-time substitution approaches are used instead, generally by defining buffers surrounding the urban area to serve as a rural reference (Q. Yang et al., 2023). This is because considering regions farther away (say, the other side of Lake Michigan) as a background reference would result in differences not just due to surface type, but also due to synoptic-scale processes.

All the model configurations are run five times with different initializations 12 hr apart between 12 May 2018 and 14 May 2018 and the simulations are run till 1 September 2018 (J. Wang et al., 2023). The five different initializations form an ensemble and are meant to test the sensitivity of the results to initial conditions. The results for the 3 months of summer (June, July, August) are used in the study. The WRF outputs are at hourly resolution for each model grid.

2.3. Calculating Heat Indices

We estimate heat indices from the model simulations to represent human physiological response to heat extremes. First, we calculate wet-bulb temperature, a thermodynamic measure of humidity in a parcel of air, which is often used as a proxy for heat stress in climate studies (Im et al., 2017; Raymond et al., 2020; Sherwood &

Huber, 2010) using an iterative approach (Stipanuk, 1973). We also compute wet bulb globe temperature, which is the ISO (International Organization for Standardization) standard for occupational heat stress (Iso, 2017), using the following empirical equation (Ono & Tonouchi, 2014):

WBGT =
$$0.735 \times AT + 0.0374 \times RH + 0.00292 \times AT \times RH + 7.619 \times SR - 4.557 \times SR^{2}$$

-0.0572 × WS - 4.064 (1)

where WBGT is in °C, SR is the solar insolation at the surface (in kW m⁻²), WS is wind speed at 10 m (in m s⁻¹), RH is the 2-m relative humidity (in %), and AT is the 2-m air temperature (in °C). Wet-bulb temperature generally has a stronger dependence on humidity than other heat stress metrics (Chakraborty et al., 2022), and is not strongly linked to health outcomes till physiological thresholds are reached (Sherwood, 2018). Moreover, these physiological thresholds assume no energy input (such as from solar radiation) into the human body and continuous gale force winds, which is quite unrealistic. Since urbanization and the lake both impact wind speed, and solar exposure is expected under most outdoor conditions during daytime, wet bulb globe temperature is a more complete indicator of outdoor heat stress (Heo et al., 2019), which we focus on when describing the results. The overall patterns found in the study are similar when using wet-bulb temperature, with this variable showing even weaker associations with distance from the coast and income than wet bulb globe temperature (Figures S1, S3, and S4 in Supporting Information S1).

2.4. Data Processing at Neighborhood-Scale

Average diurnal cycles are generated for all relevant variables (air temperature, relative humidity, wet bulb globe temperature, wet-bulb temperature, and skin temperature) from the hourly model outputs for summer 2018. From these cycles, the highest and lowest values are extracted for each grid, representing the maximum and minimum value for the average summer day in 2018, and summarized for each of the community areas. Similarly, the 95th and 98th percentile of these variables are also extracted from the hourly outputs and combined to create summaries for each geographic aggregation.

To check the consistency of patterns with observations, we also extract satellite data for summer 2018 over Chicago and summarize for the same regions. Satellite-derived skin temperature (more commonly called land surface temperature; LST) is from 8-day composite 1 km MODIS observations on board the Aqua satellite. Pixel-level quality control flags are used to only keep data with an uncertainty of less than or equal to 3°C, following previous studies (Chakraborty et al., 2020). Note that this uncertainty relates to the algorithm used to estimate LST from MODIS observations in the thermal bands; and is different from the sources of uncertainty in the coupled model simulations (see Sections 2.7 and 3.6). The Aqua overpass for day and night are at around 1:30 p.m. and 1:30 a.m. local time, respectively, which is not the same as the time of the maximum and minimum average skin temperature. For a more accurate comparison, WRF-simulated skin temperatures corresponding to these overpasses (1–2 p.m. local time for daytime; 1–2 a.m. for nighttime) are estimated from the control runs (Figure 1; Figures S1 and S2 in Supporting Information S1). We also calculate the normalized difference vegetation index (NDVI), a proxy of live green vegetation at the surface (Rouse et al., 1974), from MODIS surface reflectance product. The NDVI is calculated as:

$$NDVI = (NIR - RED)/(NIR + RED)$$
(2)

where NIR is the reflectance in the near-infrared and RED is the reflectance in the red band. Only the best quality pixels are selected before calculating the NDVI. For both skin temperature and NDVI, we also mask out all pixels corresponding to surface water (at 30 m resolution) from the Global Surface Water data set (Pekel et al., 2016) since the urban grids in WRF do not include any surface water fraction. All satellite remote sensing and geospatial data processing are done on the Google Earth Engine platform (Gorelick et al., 2017).

2.5. Factor Analysis Using Gridded Model Simulations

We examine the impact of different factors in Equation 1 on spatial variability of maximum and minimum average wet bulb globe temperature. For this, the factors (AT, RH, WS, and SR) corresponding to the time of these maximum and minimum average WBGT values are extracted for each grid overlaying Chicago from the control simulations. Then, multiple linear regressions are used to express wet bulb globe temperature (maximum or minimum average) as a function of its factors:

WBGT = $\beta_0 + \beta_1 AT + \beta_2 RH + \beta_3 WS + \beta_4 SR$

24711

(3)

where β_0 is the intercept and β_1 , β_2 , β_3 , and β_4 are the regression coefficients of air temperature, relative humidity, wind speed, and solar radiation, respectively. These coefficients give the sensitivity of wet bulb globe temperature to a unit change in the associated factor, assuming other factors are held constant. Overall, the magnitude and direction of these coefficients give an estimate of the strength and direction of association between the factor and wet bulb globe temperature. Since the factors have different range of values, which would impact the regression coefficients, we provide a second perspective by also computing similar sensitivities after rescaling all factors to lie between 0 and 1. In this scaled case, the relative importance of factors estimated purely due to the unit used is accounted for. Note that this method assumes linear independence of the factors. In reality, some of these factors are not independent. For instance, increasing air temperature would reduce relative humidity even if the total moisture content of the air remains constant. This potential collinearity between the factors should be kept in mind when assessing the results. However, that air temperature is vastly more important for wet bulb globe temperature is seen in the regression coefficients found here (see Section 3) and also easily interpretable from the coefficients in Equation 1.

2.6. Extracting Crowdsourced Weather Data in Chicago

The model configuration used here have been evaluated against buoy measurements and in situ measurements throughout the Great Lakes Region in a previous study (J. Wang et al., 2023). However, in that study, the focus was more on day-night patterns and overall accuracy rather than spatial variability within Chicago. Since urban areas rarely have standard weather stations, our primary evaluation of the modeled skin temperature uses satel-lite observations. However, with the intent of capturing both the magnitude and the spatial variability of other relevant variables important for heat stress, we also extract all citizen science weather station data in Chicago that have data covering the full period of summer, 2018. These data are measured by outdoor personal weather stations from Netatmo, a manufacturer of smart home devices (Venter et al., 2021). Although only five such stations exist within Chicago (Figure S3a in Supporting Information S1), they are more spread out than the airport weather station data normally used for such model evaluations (Tan et al., 2022). We calculate the maximum and minimum average air temperature and relative humidity from each of these stations for summer 2018 and examine associations between these values and the corresponding neighborhood-level estimates from the five-member ensemble mean of the WRF control simulations.

2.7. Model Evaluation and Uncertainties

Though we evaluate our WRF-simulated variables against satellites observations after aggregating to the neighborhood scale (see below and Figures S1 and S2 in Supporting Information S1), against both buoy and in situ measurements to examine diurnal patterns (J. Wang et al., 2023), and against crowdsourced measurements to test for spatial variability (see below and Figure S3 in Supporting Information S1), it is important to stress the uncertainties in both our model estimates and the evaluations. On the modeling side, WRF cannot fully capture the spatial variability of a heterogenous city like Chicago (Qian et al., 2022). For instance, the urban density classes have fixed radiative and thermodynamic parameters. Moreover, street-level urban vegetation is not explicitly represented, which can modulate wind speed, air temperature, and surface moisture budget, especially evapotranspiration and thus relative humidity (Krayenhoff et al., 2020). Urban hydrology is also poorly represented in WRF, with no real subsurface drainage, which would also impact relative humidity. On the evaluation side, while WRF-simulated skin temperature is integrated over an idealized 3D urban canyon, satellites provide a 2D directional view of a 3D heterogeneous urban landscape for clear-sky conditions (Du et al., 2023; Stewart et al., 2021). Thus, some differences in magnitude between skin temperatures from MODIS and WRF are expected, especially for grids with higher urban density. Finally, the citizen weather station measurements represent information about a much smaller footprint than the WRF grids.

With these uncertainties in mind, the WRF-simulated and MODIS-observed daytime estimates show a correlation of 0.29, mean bias error of -2.4° C, and a slope of the line of best fit close to unity (Figure S2a in Supporting Information S1). The impact of the vegetation representation in WRF is also evident during daytime, as the difference between modeled and observed skin temperature is positively correlated with normalized difference vegetation index (NDVI; Figure S2c in Supporting Information S1), which is a proxy for live green vegetation at the surface (Rouse et al., 1974). At night, however, the model cannot capture the variability of observed skin temperature across neighborhoods, showing a negative correlation with an r^2 value close to 0 (Figure S2b in Supporting Information S1). This discrepancy between modeled and observed nighttime skin temperature is also evident from the opposite direction of the lake-to-land gradient in Figure 4 and the opposite associations with median income (Figure 7b; Figure S8 in Supporting Information S1). In addition to differences caused by the satellite view of the 3D urban landscape providing a directional estimate of clear-sky skin temperature (unlike the WRF simulated value, which is integrated over the urban facades and accounts for all sky conditions), we note that satellite-derived values have a more continuous distribution than the simulated values (Figure S2a in Supporting Information S1). This makes sense since the skin temperature is strongly constrained by the surface energy budget, which, in turn, is constrained by the prescribed parameters of urban surfaces. These parameters being fixed in WRF (Chakraborty et al., 2021; Z.-H. Wang et al., 2011), the spatial variability of simulated skin temperature is also similarly limited, leading to the distinct clusters seen in Figure S2a of the Supporting Information S1 during daytime. This issue is of less importance for air temperature, which we further evaluate below, since it is well-mixed through horizontal advection.

While the sample size is small, for evaluations of simulated maximum and minimum average air temperature and relative humidity against weather station measurements, positive correlations are seen in all cases other than for minimum average air temperature (Figure S3 in Supporting Information S1). For air temperature, the mean bias errors are less than 0.5°C, while the errors are higher for relative humidity. Minimum average relative humidity, which is during daytime, is largely underestimated by WRF even though the general spatial pattern is captured. This is potentially also related to the lack of explicit vegetation representation and hydrology within the urban grids, which would impact the near-surface relative humidity simulated by the model. In comparison, the personal weather stations are generally not set up over impervious surfaces, but rather in lawns and backyards, which would have higher relative humidity. Finally, moisture emissions from anthropogenic activities are also poorly represented in models (X. Huang et al., 2021; Z. Wang et al., 2021), which would also impact relative humidity. We stress that these evaluations are done more as a sanity check than to confirm the "truth" represented by the model simulations. See the relevant discussions about the uncertainties earlier. However, the model simulations are helpful for capturing the relative changes among the different variables (air temperature, skin temperature, wet-bulb temperature, and wet bulb globe temperature) to perturbations like the presence of the lake or built-up areas since many of the processes modulating these variables are explicitly simulated via the model physics.

3. Results

3.1. Urban Versus Lake Impacts on Heat Stress Across Neighborhoods

Disaggregating the WRF model simulations (control runs that include both urban and lake grids) into five-member ensemble mean summaries for Chicago neighborhoods, we find a lake-to-land gradient in skin temperature, air temperature, and relative humidity, particularly during daytime (Figures 1–3; Figures S4 and S5 in Supporting Information S1). For instance, summertime maximum average air temperature and minimum average relative humidity, both during daytime, increase and decrease, respectively, as we move away from the lake shore (Figure 2a; Figure S5d in Supporting Information S1). Maximum average relative humidity also shows a lake-to-land gradient, though minimum average air temperature has a weaker gradient (Figure 2a). Wet bulb globe temperature also shows a similar but weaker lake-to-land gradient (Figure 1b).

The urban and lake impacts on these variables are estimated through perturbation simulations where the built-up and lake grids are replaced with cropland (see Section 2). With some exceptions, air temperature and wet bulb globe temperature increase due to urbanization across neighborhoods. The urban impact is particularly evident at night for these variables. The stronger urban impact on minimum air temperature compared to maximum is in line with modeling and observational estimates across cities (Qian et al., 2022; Sarangi et al., 2021; Venter et al., 2021). Urbanization appears to slightly reduce maximum average air and wet bulb globe temperature in the Near West Side, Near North Side, and the Loop neighborhoods (Figures 1a, 2b, and 2e). Urbanization also shows mixed impact on skin temperature according to the WRF simulations (Figures S4b and S5b in Supporting Information S1). Here, it should be noted that the surface of a multi-layer urban canopy is fundamentally different from that of a cropland grid, which is treated as a flat slab in the land-surface model, which could lead to some of these discrepancies. Finally, urbanization also reduces maximum and minimum relative humidity across most neighborhoods (Figures S1d and S5d in Supporting Information S1). The impact of Lake Michigan on the micro-climate of Chicago neighborhoods is more consistent, with general decreases in maximum and minimum temperature and heat indices (but with different magnitudes; see Section 3.2) and increases in relative humidity (Figures 2 and 3; Figures S4 and S5 in Supporting Information S1).





Figure 2. Urban and lake impacts on maximum air and wet-bulb globe temperature. Neighborhood-scale summaries of WRF simulated maximum average (a) air temperature and (d) wet-bulb globe temperature for summer 2018 in Chicago. (b and e) Urban impact on maximum average air temperature and wet-bulb globe temperature, respectively. (c and f) Lake impact on those variables.

3.2. Lake-To-Land Gradients in Different Measures of Heat Exposure

To quantitatively assess the lake-to-land gradients, visually evident in Figure 2, we examine associations between the variables for each neighborhood and the distance of the centroid of the neighborhood from the lake shore (Figure 4). This distance is a bulk proxy of the lake effect, including the impact of horizontal exchange of heat and moisture via lake and land breezes. We also compare the lake-to-land gradients of the modeled variables (Figure 1b) with corresponding gradients of skin temperature measured by the Moderate Resolution Imaging Spectroradiometer (MODIS) sensor aboard NASA's Aqua satellite (Figure 4a). In Figure 4, the maximum and minimum values for the satellite estimates correspond to the satellite overpass times for daytime and nighttime (\sim 1:30 p.m. and \sim 1:30 a.m. local time equatorial overpass), respectively, while in WRF, they are the actual modeled extrema (see Section 2). Although these associations can be non-linear, with stronger gradients seen at the land-water interface, we use linear models for easier interpretability of the results.

Overall, the maximum skin temperature in both the control model simulation and the MODIS observations show positive correlations with the distance from the shore with similar slopes (0.43 vs. 0.34; Figures 4a and 4b; all in °C per km), though the coefficient of determination is weaker for WRF ($r^2 = 0.24$ vs. 0.53 for MODIS). Note that these slopes are even closer (0.43 vs. 0.48; see Figure 1b) when WRF simulations corresponding to the MODIS overpass times are used (Figure 1b). For minimum (~1:30 a.m. for MODIS) skin temperature, the data are generally uncorrelated. Although the maximum air and wet bulb globe temperatures also show positive slopes with the distance from the shore, the sensitivity decreases from skin to air to wet bulb globe temperature (0.34, 0.19, and 0.07, respectively).

The counterfactual scenario where the Chicago urban grids are removed has a large impact on the gradient of skin temperature (0.34–0.05), a smaller impact on air temperature (0.19–0.13), and practically no impact on wet bulb



10.1029/2023GH000869



Figure 3. Urban and lake impacts on minimum air and wet-bulb globe temperature. Neighborhood-scale summaries of WRF simulated minimum average (a) air temperature and (d) wet-bulb globe temperature for summer 2018 in Chicago. (b and e) Urban impact on minimum average air temperature and wet-bulb globe temperature, respectively. (c and f) Lake impact on those variables.

globe temperature (0.07 for both and similar offsets; Figure 4d). The sign of the slope flips to positive from negative for maximum relative humidity. During nighttime (i.e., minimum values for all variables and maximum for relative humidity), the impact of urbanization is more evident, with the lake-to-land gradient strengthening for all cases. This suggests that the presence of Chicago counteracts the naturally expected lake-to-land gradient during nighttime. For the counterfactual scenario with no Lake Michigan, little impact is seen on minimum average variables compared to the control runs, but we see large offsets during daytime, with increases in air temperature and wet bulb globe temperature. Overall, it is evident that Lake Michigan has a stronger impact on heat exposure and heat stress in the region during daytime, while urbanization has a stronger impact at night. For almost all variables, the standard errors across the five ensemble members are greatest closest to the shore (Figure 4; Figure S6 in Supporting Information S1), demonstrating the high sensitivity of the micro-climate at the land-water interface simulated by the model to initial conditions (J. Wang et al., 2022).

3.3. Urban and Lake Impacts on Extremes Versus the Summer Climatological Mean

The previous analysis is indicative of the different sensitivities of the different measures of heat exposure and heat stress (air temperature and wet bulb globe temperature) to the distance from the water. However, these variables have different standard ranges, making the slopes not directly comparable. So, we calculate the absolute and percentage changes in the relevant variables due to urbanization and the lake within Chicago. Overall, the results seen in the sensitivity analysis are confirmed here (Table 1). Both percentage and absolute changes are higher for maximum average skin temperature than for maximum average air temperature during daytime and these changes are higher still than wet bulb globe temperature. The higher impact of urbanization on minimum average





Figure 4. Lake-to-land gradients in measures of heat exposure. Associations between (a) MODIS Aqua daytime and nighttime skin temperature (more accurately, land surface temperature) and WRF simulated maximum and minimum average (b) skin temperature, (c) air temperature, and (d) wet bulb globe temperature in summer 2018 for 77 neighborhoods in Chicago and the corresponding distance of the centroid of each neighborhood to the lake shore. The points represent the mean values from the five members of the ensemble and the error bars correspond to standard errors across the members. The lines of best fit are shown and the associated equations, coefficients of determination, and *p* values are in the legend. The square symbols show the maximum averages (daytime for MODIS) and the triangles are for the minimum averages (nighttime for MODIS). The three colors (red, blue, and green) show the results for control, no lake, and no urban simulations, respectively.

air temperature (2.02°C; compared to 0.99°C for maximum) and maximum average skin temperature (2.55°C; compared to 1.24°C for minimum) is also captured. The higher urban impact on daytime skin temperature partly relates to the lack of evaporative dissipation of heat over pervious surfaces (Paschalis et al., 2021), while this additional stored heat gets gradually released at night into the near-surface air (Zhao et al., 2014). Finally, Lake

Table 1

Impacts of Urbanization and Lake Michigan Within Chicago

	Skin temperature (°C)	Air temperature (°C)	Wet bulb globe temperature (°C)	Relative humidity (%)
	Summer composite maximum			
Urban Effect	2.55 (7.6%)	0.99 (3.7%)	0.08 (0.33%)	-7.93 (-9.13%)
Lake Effect	-0.98 (-2.63%)	-2.75 (-9%)	-0.86 (-3.39%)	5.66 (7.72%)
	Summer composite minimum			
Urban Effect	1.24 (7.17%)	2.02 (10.96%)	1.57 (9.1%)	-7.23 (-13.9%)
Lake Effect	-0.68 (-3.56%)	-0.81 (-3.82%)	-0.23 (-1.2%)	6.27 (15.08%)
	Summer 98th percentile			
Urban Effect	2.86 (7.09%)	1.84 (5.45%)	0.06 (0.2%)	-3.2 (-3.29%)
Lake Effect	-0.78 (-1.77%)	0.36 (-1%)	-0.12 (-0.41%)	1.91 (2.08%)
	Summer 95th percentile			
Urban Effect	2.34 (6.13%)	2.29 (7.18%)	0.23 (0.81%)	-3.42 (-3.62%)
Lake Effect	-0.41 (-3.32%)	-0.58 (-1.66%)	-0.26 (-0.89%)	2.91 (3.3%)

Note. Summary of urban and lake effect on maximum and minimum averages and the 95th and 98th percentiles of skin temperature, air temperature, wet bulb globe temperature, and relative humidity in summer 2018 based on the five-member ensemble mean of WRF models simulations.

Michigan generally has a stronger impact on the measures of heat exposure during daytime (up to 9% reduction for maximum average air temperature) and weaker effects at night (less than 4% decrease in all cases).

We also examine changes (absolute and percentage) in the 95th and 98th percentile of these variables (from all hourly data) for the summer of 2018 due to the lake and urbanization (Table 1). Although the urban impact on the 95th and 98th percentile of air temperature is higher than that for the maximum average, the changes in wet bulb globe temperature extremes are lower. This is possible because as the air warms, all else remaining constant, the relative humidity will decrease, thus moderating heat stress. Considering the whole of Chicago, the 95th and 98th percentiles of wet bulb globe temperature (28.6 and 29.8°C, respectively) exceed the high risk threshold (28°C) used by the U.S. military (Willett & Sherwood, 2012). However, we find a low impact of urbanization on summer heat stress extremes (upper bound increase of 0.23°C for the 95th percentile of summer wet bulb globe temperature). Finally, both urbanization and the lake have smaller effects on the 98th percentile of heat exposure and heat stress than on the 95th percentiles, suggesting a flattening of the upper tail of their distributions. It is important to note that our goal here was to examine the climatological extremes for these variables during a typical summer. Individual extreme heat events are usually much shorter (Tan et al., 2022) and influenced by regional and continental-scale synoptic processes during or before those periods.

3.4. Changes in Other Factors and Contributors to Heat Stress

In addition to air temperature and relative humidity, wet bulb globe temperature is also a function of wind speed and solar insolation (see Equation 1 in Section 2). Urbanization can impact both of these variables. We examine diurnal median composites of wind speed and the shortwave radiation incident at the surface over the city from one of the members of the ensemble (Figure 5). As expected, wind speed peaks during daytime, with large spread across the grids overlaying Chicago, while nights are consistently calmer across these grids (Figure 5a). Overall, removing the urban grids increases median wind speed, particularly at night, which is expected due to the decreased surface roughness (from urban structures to cropland) (Qian et al., 2022; Zhao et al., 2014). During daytime, wind speed changes are less. This does not mean that the urban impacts are less. Instead, it is because the land-to-lake temperature gradients are decreased, which weakens the lake breeze. This effect partly compensates for the impeding effects of urban roughness. The role of both roughness and thermal gradients is also evident from the almost uniformly low wind speed throughout the day in the no lake simulation, when urban roughness effects are still active, and the lake (or rather, the croplands where the lake would have been)-to-land temperature differences are small (Figure 4b).

Incoming shortwave does not vary much across grids and shows minor changes across these simulations, with lowest value in the no lake run (Figure 5b). This is probably because of changes in cloud cover in the three simulations influenced by changes in surface properties and moisture fluxes (J. Wang et al., 2022). We also look at the diurnal evolution of the boundary layer height across these grids for the three simulations (Figure 5c). This height is a proxy for the near-surface stability and overall convective efficiency over the city. Overall, we see low boundary layer height during stable nights, with large increases during daytime as the atmosphere becomes unstable. The boundary layer is highest in the no lake simulation, the inverse of what is seen for wind speed. An interesting feature is the growth of boundary layer after around 7 p.m. local time in both simulations with the multilayer urban canopy model. This is a consequence of the sensitivity of this boundary layer scheme to the prescribed addition of anthropogenic flux during that time. This bump disappears when a different boundary layer scheme is used (not shown).

Using multiple linear regressions, we examine relative importance of each factor considered when calculating the wet bulb globe temperature on the spatial variability of its maximum average and minimum average values over the Chicago grids. The linear models can almost perfectly capture the variability in wet bulb globe temperature, with adjusted R^2 of around 0.99 and 0.97 for the models of maximum and minimum wet bulb globe temperature, respectively. The relative importance is given by the regression coefficients, with the sign of the coefficients showing the direction of the associations (Figure 6). The signs are consistent with Equation 1 of the Methods, with wet bulb globe temperature increasing with air temperature, relative humidity, and solar radiation, and decreasing with wind speed. Across members of the ensemble, air temperature has the greatest role in controlling the spatial variability of both maximum and minimum average wet bulb globe temperature over the city at this scale. The second most important variable depends on whether the variables are scaled before the regression coefficients are computed. For unscaled variables, wind speed is most important, while relative humidity becomes more important when the variables are scaled due to its much larger magnitude compared to wind speed. Solar radiation is generally the least importance, except for minimum average wet bulb globe temperature, which may





Figure 5. Diurnal variability of other relevant factors. Diurnal composites of (a) wind speed, (b) incoming shortwave radiation, and c boundary layer height in summer 2018 across model grids overlaying Chicago. The upper and lower lines represent the 75% and 25% percentile (across grids) of the mean values, and the middle line is for the median, all by hour of the day. Results are shown for the first member of the ensemble of WRF simulations.

seem counterintuitive, but is because the minimum wet-bulb temperature can be right at dawn for some grids with some non-zero shortwave, which becomes relatively more important when the variables are scaled.

3.5. Income-Based Disparities in Different Metrics of Heat Exposure

A reason for using neighborhood-scale summaries from the model simulations is to match them to socioeconomic data collected for the same geographic regions. Several studies have examined disparities in heat exposure in US cities using similar methodology, but focusing on satellite-derived skin temperature (Chakraborty et al., 2020; Hoffman et al., 2020; Hsu et al., 2021). Since skin temperature is less relevant to physiological response to heat than heat indices that combine multiple factors, we examine associations between skin temperature, air temperature, and wet bulb globe temperature against the median income per capita for Chicago (Figure 7). Overall, we see negative and statistically significant (p < 0.05) relationships between income and the maximum



Figure 6. Sensitivity of wet bulb globe temperature to factors. Regression coefficients for air temperature (AT), relative humidity (RH), wind speed (WS), and incoming shortwave radiation (SR) from multiple linear regressions that explain spatial variability of (a) maximum and (b) minimum wet bulb globe temperature (WBGT) over the model grid overlaying Chicago. The error bars show the standard error across the five members of the ensemble. (c and d) are similar to a and b, but from regressions where all variables have been scaled to lie between 0 and 1.

summer averages of all these variables, suggesting lower peak heat exposure in richer Chicago neighborhoods (Figure 7a). However, the slope of these relationships is strongest for skin temperature (-1.29° C per \$10,000 higher median income per capita) and weakest for wet bulb globe temperature (-0.23° C per \$10,000 higher median income per capita). This makes conceptual sense since air and wet bulb globe temperature are strongly affected by advection and the latter is also influenced by urbanization-induced drying (Chakraborty et al., 2022). Overall, we should be cautious about quantitative estimates of disparities in heat exposure based on skin temperature showing a positive association with income and wet bulb globe temperature showing almost no association (Figure 7b). The nighttime associations with income are also not entirely consistent across model configurations, which partly relate to how anthropogenic heat is prescribed in WRF (see Section 3.6).

3.6. Confirming Key Results With Different Model Configurations

We chose our default model configuration based on previous work done over the Great Lakes region, which was focused on improving prescribed boundary conditions for lake surface temperature and capturing near-surface climate over the lakes (J. Wang et al., 2022). Due to the sensitivity of urban climate to model complexity (Qian et al., 2022), we consider a couple of other configurations and run them for the control simulation. First, we run a five-member ensemble using the same configuration but using only the Noah LSM instead of BEP and BEM. Without the multi-layer urban canopy, the urban land is treated as a slab with modified surface properties with no explicit building impact on airflow or energy exchange. We also run a three-member ensemble with BEP and BEM, but using the Mellor–Yamada–Janjic (MYJ) scheme (Janjić, 1994), another commonly used boundary layer scheme in mesoscale models, instead of the YSU scheme. Finally, since the 4 km model grid is somewhat coarse, being similar in area to the community areas, we also set up a three-way nested domain (12–4 km–1.333 km) that can simulate finer-scale variability to examine consistency of our results, especially relevant for the disparity analysis using neighborhood-scale socioeconomic data. Only one ensemble member or simulation is done for the nested case due to computational bottlenecks for a full summer simulation.





Figure 7. Income-based disparities in different measures of heat exposure. Associations between WRF simulated (a) maximum and (b) minimum average skin temperature, air temperature, and wet bulb globe temperature in summer 2018 for 77 neighborhoods in Chicago and the corresponding median income per capita. The points represent the mean values from the five members of the ensemble and the error bars correspond to standard errors across the members. The lines of best fit are shown and the associated equations, coefficients of determination, and p values are in the legend.

We consider the lake-to-land gradients in maximum and minimum average skin temperature, air temperature, and wet bulb globe temperature and the associated correlations with median income as the standard results for comparison. In all cases, the maximum average skin temperature, air temperature, and wet bulb globe temperature show a positive correlation with distance from the lake shore, a proxy for the combined impact of the lake and urbanization. More importantly, similar to our main results (Figures 1b and 4), the gradient is steeper for skin temperature than for air temperature, while wet bulb globe temperature shows the least gradient (Figure 8). Although different configurations show different magnitudes of the gradients, which is a function of both model configuration and the internal model variability in a coupled framework, the stronger sensitivity of skin temperature is consistently seen. Similarly, disparities in maximum heat exposure regardless of the measure used are seen across configurations (Figure 9). However, the magnitudes of the disparities are higher for skin temperature than for air temperature shows the least disparities are higher for skin temperature than for air temperature in maximum heat exposure regardless of the measure used are seen across configurations (Figure 9). However, the magnitudes of the disparities are higher for skin temperature than for air temperature shows the least disparities are higher for skin temperature than for air temperature shows the least disparities are higher for skin temperature than for air temperature shows the least disparities are higher for skin temperature than for air temperature shows the least disparities are higher for skin temperature than for air temperature, and wet bulb globe temperature shows the least disparity.

For minimum average values, the results are mixed. For lake-to-land gradients, the sensitivities are slightly negative or close to zero in our default runs as well as using the MYJ scheme and the three-nested simulation (Figure 4; Figures S9b and S9c in Supporting Information S1). But the Noah LSM shows positive associations with the distance from the lake shore. This could be because of lack of explicit building effects in Noah LSM compared to the other configurations. Note, however, that the negligible gradient in wet bulb globe temperature is still captured by this configuration (Figure S9a in Supporting Information S1). Overall, for daytime, when heat stress would be maximum, our results show qualitatively consistent results regardless of model setup, physics schemes, and spatial resolutions, with similar lake-to-land gradients and associations with median income (Figures 1b, 4, 7, 8, 9; Figures S9 and S10 in Supporting Information S1). Moreover, air temperature and heat stress show lower spatial variability than skin temperature, which makes conceptual sense and is in line with previous observational and modeling estimates at various scales (Chakraborty, Newman, et al., 2023; Chakraborty et al., 2022; Ho et al., 2016; Venter et al., 2021). However, we did find that the models configurations can give conflicting results for disparities in minimum average measures of heat exposure (Figure 7; Figure S10 in Supporting Information S1), which requires better representation of nighttime urban processes in the model.

10.1029/2023GH000869





Figure 8. Comparing lake-to-land gradients in maximum average of the variables across model configurations. Associations between maximum average skin temperature, air temperature, and wet bulb globe temperature in summer 2018 for Chicago neighborhoods and the corresponding distance of the centroid of each community area to the lake shore from WRF simulations with (a) NOAH land surface model, (b) MYJ boundary layer scheme, and (c) three-way nesting. The points represent the mean values (five-member ensemble for (a) and three-member ensemble for (b) and the error bars correspond to standard errors across the members when they are used. The lines of best fit are shown and the associated equations, coefficients of determination, and p values are in the legend.





4. Discussion

The majority of the global human population lives near water bodies, whether in coastal areas or along freshwater shorelines (Crowell et al., 2007; Kummu et al., 2011). Similarly, large cities have frequently developed along the waterfront, with 14 of 17 global megacities (population >10 million) and around 40% of major cities (population between 1 and 10 million) being in coastal areas (Tibbetts, 2002). In 2010, ~40% of the US population lived in coastal shoreline counties (adjacent to the open ocean, major estuaries, and the Great Lakes), which had disproportionate long-term (1970–2010) area-adjusted population growth (~3.5 times the nationwide average) (NOAA National Centers for Environmental Information, Climate at a Glance: Global Time Series, Published October 2021, Retrieved on 11 November 2021). These coastal areas will be major centers of future climate change impacts due to continuous population growth and vulnerability to flood risk, sea level rise, heat stress, and other extreme weather phenomena (Diffenbaugh et al., 2007; Hallegatte et al., 2013; Yin et al., 2009). In Chicago, our model simulations show that the presence of Lake Michigan reduces maximum and minimum heat exposure and heat stress over most neighborhoods during summer, with the effects generally strongest in neighborhoods adjacent to the lake shore. In contrast, urbanization increases most metrics of heat exposure, particularly during nighttime, with these increases intensifying with distance from the lake shore. However, both the model and satellite observations show stronger gradients of skin temperature away from the lake shore than of simulated air and wet bulb globe temperature. This makes sense because air and wet bulb globe temperatures are well-mixed, reducing thermal gradients compared to the surface, which is strongly constrained by the local surface energy budget. Similarly, the impact of urbanization on maximum average wet bulb globe temperature (Figure 2e), which is a standard metric of heat stress linked to health outcomes, is much lower than that on skin temperature. This is partly due to the compensating effect of humidity on heat stress (Chakraborty et al., 2022; Sarangi et al., 2021), which (humidity) is higher near the lake shore. The reduced variability (compared to skin and air temperature) seen for wet bulb globe temperature is also found for wet-bulb temperature (Figures S4-S7 in Supporting Information S1; see Section 2), a thermodynamic measure of the moisture content of air, often used in the Earth sciences as a proxy for moist heat stress (Im et al., 2017; Raymond et al., 2020; Sherwood & Huber, 2010). These results are important for contextualizing the usefulness of open water for alleviating heat stress (Theeuwes et al., 2013), and the relevance of quantitative estimates of heat reduction and variability in heat exposure using satellite-derived skin temperature (Y. Chen et al., 2022; He et al., 2019; Manoli et al., 2019). Overall, studies examining public health implications should be careful when providing magnitude of changes for skin temperature, since these changes are not equivalent to changes in physiological metrics of heat stress (Chakraborty et al., 2022; Li et al., 2023; Turner et al., 2022).

Similarly, when examining disparities in different metrics of heat exposure across neighborhoods in Chicago, we find large differences in the sensitivities to median income. Skin temperature generally shows the strongest sensitivity to income, while wet bulb globe temperature shows the weakest association. This result is replicated when we look at associations between these variables (including wet-bulb temperature) and the Hardship index, a more comprehensive metric for structural urban inequity (Amdat, 2021) (Figure S7a in Supporting Information S1; also see Section 2). The maximum summer averages of all variables generally increase with Hardship index, with the greatest sensitivity for skin temperature and the least for wet-bulb temperature (Figure S7 in Supporting Information S1). These results indicate the importance of examining disparities in physiologically relevant estimates of heat stress instead of satellite-derived skin temperature to accurately quantify urban environmental inequities, which has also been noted in a nationwide study using a much simpler urban modeling framework (Chakraborty, Newman, et al., 2023). Although our study does show that using skin temperature, as done frequently in previous studies, may exaggerate the magnitude of disparities in heat exposure, it is important to stress that public health consequences of weather extremes depend on both exposure and vulnerability (Hsu et al., 2021). For heat stress, poorer populations are more vulnerable even when the exposure is identical. For instance, a working air conditioner, generally less available to lower-income populations (Romitti et al., 2022), was the strongest protective factor for heat-related death during the 1999 Chicago heat wave (Naughton et al., 2002).

Although the variability in air temperature due to Lake Michigan is consistent with previous estimates (Conry et al., 2015; J. Yang et al., 2022), the interactions between the lake and urbanization on heat stress and its disparities in Chicago are influenced by air temperature, relative humidity, and wind, with a distinct overall response for seasonal extremes (Table 1). These complexities makes it more difficult to predict future heat stress in Chicago and other coastal cities, with multiple factors coming into play, including additional drying and warming due to

urban expansion and increased onshore penetration by lake breezes (Conry et al., 2015). Accurately capturing these competing mechanisms requires not just better representations of cities and lakes in models, but also reasonable estimates of dynamic changes in land cover, including urban greening and irrigation practices, which are still poorly constrained (Krayenhoff et al., 2020; Qian et al., 2020, 2022; Sharma et al., 2020; J. Wang et al., 2022), with currently unknown overall impacts on urban heat stress (Im et al., 2017). As we head toward a warmer, wetter, and more urbanized future (Lewis & Maslin, 2015; W. Wang et al., 2021), it is critical to continue to build these tools that can provide accurate and actionable data for urban climate adaptation and mitigation, especially to help vulnerable waterfront communities (Sharma et al., 2020).

5. Conclusion

We examine the role of urbanization and Lake Michigan on different measures of heat exposure in Chicago using ensembles of weather model simulations. Urbanization increases heat exposure, while Lake Michigan decreases it. However, the degree of urban and lake impact on heat exposure depends on the metric or variable used. Skin temperature generally responds the most to urban and lake effects, while a physiologically relevant metric of heat stress like wet bulb globe temperature responds the least. This is because of the compensating impacts of humidity on wet bulb globe temperature. Overall, the competing mechanisms mean that lake-to-land gradients in maximum wet bulb globe temperature is much weaker than well-known gradients for air and skin temperature. Consequently, potential disparities in maximum ambient heat stress are less than what would be seen from satellite observations or in air temperature alone. Our results demonstrate the importance of using appropriate metrics of heat stress to provide relevant quantitative estimates from urban heat monitoring efforts and for informing urban adaptation and mitigation strategies.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The spatial polygons for the community areas and the socioeconomic data were accessed through the Chicago Data Portal (City of Chicago, 2014). All neighborhood-scale summaries generated in this study can be found here (Chakraborty, Wang, et al., 2023). *Code availability*: The WRF model code is open source and can be accessed at https://github.com/wrf-model/WRF. The Google Earth Engine scripts for summarizing WRF simulations and satellite observations can be found through Zenodo (Chakraborty, Wang, et al., 2023).

References

Amdat, W. C. (2021). The Chicago hardship index: An introduction to urban inequity. *Journal of Statistics and Data Science Education*, 1–9.

Anderson, G. B., Bell, M. L., & Peng, R. D. (2013). Methods to calculate the heat index as an exposure metric in environmental health research. *Environmental Health Perspectives*, 121(10), 1111–1119. https://doi.org/10.1289/ehp.1206273

Benz, S. A., & Burney, J. A. (2021). Widespread race and class disparities in surface urban heat extremes across the United States. *Earth's Future*, 9(7). https://doi.org/10.1029/2021EF002016

- Birch, C. E., Roberts, M. J., Garcia-Carreras, L., Ackerley, D., Reeder, M. J., Lock, A. P., & Schiemann, R. (2015). Sea-breeze dynamics and convection initiation: The influence of convective parameterization in weather and climate model biases. *Journal of Climate*, 28(20), 8093– 8108. https://doi.org/10.1175/jcli-d-14-00850.1
- Chakraborty, T., Hsu, A., Manya, D., & Sheriff, G. (2019). Disproportionately higher exposure to urban heat in lower-income neighborhoods: A multi-city perspective. *Environmental Research Letters*, 14(10), 105003. https://doi.org/10.1088/1748-9326/ab3b99
- Chakraborty, T., Hsu, A., Manya, D., & Sheriff, G. (2020). A spatially explicit surface urban heat island database for the United States: Characterization, uncertainties, and possible applications. *ISPRS Journal of Photogrammetry and Remote Sensing*, 168, 74–88. https://doi.org/10.1016/j. isprsjprs.2020.07.021
- Chakraborty, T., Lee, X., Ermida, S., & Zhan, W. (2021). On the land emissivity assumption and Landsat-derived surface urban heat islands: A global analysis. *Remote Sensing of Environment*, 265, 112682. https://doi.org/10.1016/j.rse.2021.112682
- Chakraborty, T., Newman, A. J., Qian, Y., Hsu, A., & Sheriff, G. (2023). Residential segregation and outdoor urban moist heat stress disparities in the United States. One Earth, 6(6), 738–750. https://doi.org/10.1016/j.oneear.2023.05.016
- Chakraborty, T., Venter, Z. S., Qian, Y., & Lee, X. (2022). Lower urban humidity moderates outdoor heat stress. AGU Advances, 3(5), e2022AV000729. https://doi.org/10.1029/2022AV000729

Chakraborty, T., Wang, J., Qian, Y., Pringle, W., Yang, Z., & Xue, P. (2023). TC25/Chicago_COMPASS: Present-day heat-related estimates for Chicago community areas [COMPASS-GLM] (COMPASS-GLM) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.8346441

Chen, F., & Dudhia, J. (2001). Coupling an advanced land surface-hydrology model with the Penn State-NCAR MM5 modeling system. Part I: Model implementation and sensitivity. *Monthly Weather Review*, 129(4), 569–585. https://doi.org/10.1175/1520-0493(2001)129<0569:caalsh>2.0.co;2

24711403, 2023, 11, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GH000869, Wiley Online Library on [23/11/2023]. See the Terms and Conditions (https://onlinelibrary.wiley

Acknowledgments

This study is supported by COMPASS-GLM, a multi-institutional project supported by the U.S. Department of Energy (DOE), Office of Science, Office of Biological and Environmental Research as part of the Regional and Global Modeling and Analysis (RGMA) program, Multi-sector Dynamics Modeling (MSD) program, and Earth System Model Development (ESMD) program, as well as a DOE Early Career Research Program Grant to T.C. This study is also Contribution No. 92 of the Great Lakes Research Center at Michigan Technological University. Computational resources are provided by the DOE-supported National Energy Research Scientific Computing Center and Argonne Leadership Computing Facility. Initial calculations were done using the NCAR Command Language (version 6.6.2; software), 2019, Boulder, CO: UCAR/NCAR/ CISL/TDD, http://dx.doi.org/10.5065/ D6WD3XH5, while further geospatial processing was done on the Google Earth Engine platform. The Pacific Northwest National Laboratory is operated for DOE by Battelle Memorial Institute under contract DE-AC05-76RL01830.

com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons Licens

- Chen, Y., Yang, J., Yang, R., Xiao, X., Xia, J., & Cecilia) (2022). Contribution of urban functional zones to the spatial distribution of urban thermal environment. *Building and Environment*, *216*, 109000. https://doi.org/10.1016/j.buildenv.2022.109000
- City of Chicago. (2014). Census data—Selected socioeconomic indicators in Chicago, 2008–2012. [Online]. Retrieved from https://data.cityofchicago.org/Health-Human-Services/Census-Data-Selected-socioeconomic-indicators-in-C/kn9c-c2s2
- Conry, P., Sharma, A., Potosnak, M. J., Leo, L. S., Bensman, E., Hellmann, J. J., & Fernando, H. J. (2015). Chicago's heat island and climate change: Bridging the scales via dynamical downscaling. *Journal of Applied Meteorology and Climatology*, 54(7), 1430–1448. https://doi. org/10.1175/jamc-d-14-0241.1
- Crowell, M., Edelman, S., Coulton, K., & McAfee, S. (2007). How many people live in coastal areas? Journal of Coastal Research, 23(5), iii–vi. https://doi.org/10.2112/07a-0017.1
- Diffenbaugh, N. S., Pal, J. S., Giorgi, F., & Gao, X. (2007). Heat stress intensification in the Mediterranean climate change hotspot. Geophysical Research Letters, 34(11), L11706. https://doi.org/10.1029/2007gl030000
- Du, H., Zhan, W., Liu, Z., Krayenhoff, E. S., Chakraborty, T., Zhao, L., et al. (2023). Global mapping of urban thermal anisotropy reveals substantial potential biases for remotely sensed urban climates. *Science Bulletin*, 68(16), 1809–1818. https://doi.org/10.1016/j.scib.2023.06.032
- Gorelick, N., Hancher, M., Dixon, M., Ilyushchenko, S., Thau, D., & Moore, R. (2017). Google Earth engine: Planetary-scale geospatial analysis for everyone. *Remote Sensing of Environment*, 202, 18–27. https://doi.org/10.1016/j.rse.2017.06.031
- Hallegatte, S., Green, C., Nicholls, R. J., & Corfee-Morlot, J. (2013). Future flood losses in major coastal cities. *Nature Climate Change*, 3(9), 802–806. https://doi.org/10.1038/nclimate1979
- He, B.-J., Zhao, Z.-Q., Shen, L.-D., Wang, H.-B., & Li, L.-G. (2019). An approach to examining performances of cool/hot sources in mitigating/ enhancing land surface temperature under different temperature backgrounds based on landsat 8 image. Sustainable Cities and Society, 44, 416–427. https://doi.org/10.1016/j.scs.2018.10.049
- Heo, S., Bell, M. L., & Lee, J.-T. (2019). Comparison of health risks by heat wave definition: Applicability of wet bulb globe temperature for heat wave criteria. *Environmental Research*, 168, 158–170. https://doi.org/10.1016/j.envres.2018.09.032
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Ho, H. C., Knudby, A., Xu, Y., Hodul, M., & Aminipouri, M. (2016). A comparison of urban heat islands mapped using skin temperature, air temperature, and apparent temperature (Humidex), for the greater Vancouver area. Science of the Total Environment, 544, 929–938. https:// doi.org/10.1016/j.scitotenv.2015.12.021
- Hoffman, J. S., Shandas, V., & Pendleton, N. (2020). The effects of historical housing policies on resident exposure to intra-urban heat: A study of 108 US urban areas. *Climate*, 8(1), 12. https://doi.org/10.3390/cli8010012
- Hong, S.-Y., & Lim, J.-O. J. (2006). The WRF single-moment 6-class microphysics scheme (WSM6). Asia-Pacific Journal of Atmospheric Sciences, 42(2), 129–151.
- Hsu, A., Sheriff, G., Chakraborty, T., & Manya, D. (2021). Disproportionate exposure to urban heat island intensity across major US cities. *Nature Communications*, 12(1), 2721. https://doi.org/10.1038/s41467-021-22799-5
- Hu, X.-M., & Xue, M. (2016). Influence of synoptic sea-breeze fronts on the urban heat island intensity in Dallas–Fort Worth, Texas. Monthly Weather Review, 144(4), 1487–1507. https://doi.org/10.1175/mwr-d-15-0201.1
- Huang, K., Lee, X., Stone, B., Jr., Knievel, J., Bell, M. L., & Seto, K. C. (2021). Persistent increases in nightime heat stress from urban expansion despite heat island mitigation. *Journal of Geophysical Research: Atmospheres*, 126(4), e2020JD033831. https://doi.org/10.1029/2020jd033831
- Huang, X., Song, J., Wang, C., Chui, T. F. M., & Chan, P. W. (2021). The synergistic effect of urban heat and moisture islands in a compact highrise city. *Building and Environment*, 205, 108274. https://doi.org/10.1016/j.buildenv.2021.108274
- Iacono, M. J., Delamere, J. S., Mlawer, E. J., Shephard, M. W., Clough, S. A., & Collins, W. D. (2008). Radiative forcing by long-lived greenhouse gases: Calculations with the AER radiative transfer models. *Journal of Geophysical Research*, 113(D13), D13103. https://doi. org/10.1029/2008jd009944
- Im, E.-S., Pal, J. S., & Eltahir, E. A. B. (2017). Deadly heat waves projected in the densely populated agricultural regions of South Asia. Science Advances, 3(8), e1603322. https://doi.org/10.1126/sciadv.1603322
- Iso, B. S. (2017). 7243: Ergonomics of the thermal environment—Assessment of heat stress using the WBGT (wet bulb globe temperature) index. International Organization for Standardization.
- Janjić, Z. I. (1994). The step-mountain eta coordinate model: Further developments of the convection, viscous sublayer, and turbulence closure schemes. *Monthly Weather Review*, 122(5), 927–945. https://doi.org/10.1175/1520-0493(1994)122<0927:tsmecm>2.0.co;2
- Jiménez, P. A., Dudhia, J., González-Rouco, J. F., Navarro, J., Montávez, J. P., & García-Bustamante, E. (2012). A revised scheme for the WRF surface layer formulation. *Monthly Weather Review*, 140(3), 898–918. https://doi.org/10.1175/mwr-d-11-00056.1
- Juday, L. J. (2015). The changing shape of American cities. Charlottesville: Demographics Research Group. Weldon Cooper Center for Public Service: University of Virginia.
- Keeler, J. M., & Kristovich, D. A. (2012). Observations of urban heat island influence on lake-breeze frontal movement. Journal of Applied Meteorology and Climatology, 51(4), 702–710. https://doi.org/10.1175/jamc-d-11-0166.1
- Krayenhoff, E. S., Jiang, T., Christen, A., Martilli, A., Oke, T. R., Bailey, B. N., et al. (2020). A multi-layer urban canopy meteorological model with trees (BEP-Tree): Street tree impacts on pedestrian-level climate. Urban Climate, 32, 100590. https://doi.org/10.1016/j.uclim.2020.100590
- Kummu, M., de Moel, H., Ward, P. J., & Varis, O. (2011). How close do we live to water? A global analysis of population distance to freshwater bodies. PLoS One, 6(6), e20578. https://doi.org/10.1371/journal.pone.0020578
- Kunkel, K. E., Changnon, S. A., Reinke, B. C., & Arritt, R. W. (1996). The July 1995 heat wave in the Midwest: A climatic perspective and critical weather factors. Bulletin of the American Meteorological Society, 77(7), 1507–1518. https://doi.org/10.1175/1520-0477(1996)077<1507:tjhwit>2.0.co;2
- Laird, N. F., Kristovich, D. A., Liang, X.-Z., Arritt, R. W., & Labas, K. (2001). Lake Michigan lake breezes: Climatology, local forcing, and synoptic environment. *Journal of Applied Meteorology*, 40(3), 409–424. https://doi.org/10.1175/1520-0450(2001)040<0409:lmlbcl>2.0.co;2 Lewis, S. L., & Maslin, M. A. (2015). Defining the Anthropocene. *Nature*, 519(7542), 171–180. https://doi.org/10.1038/nature14258
- Li, X., Chakraborty, T., & Wang, G. (2023). Comparing land surface temperature and mean radiant temperature for urban heat mapping in Philadelphia. Urban Climate, 51, 101615. https://doi.org/10.1016/j.uclim.2023.101615
- Manoli, G., Fatichi, S., Schläpfer, M., Yu, K., Crowther, T. W., Meili, N., et al. (2019). Magnitude of urban heat islands largely explained by climate and population. *Nature*, 573(7772), 55–60. https://doi.org/10.1038/s41586-019-1512-9
- Martilli, A., Clappier, A., & Rotach, M. W. (2002). An urban surface exchange parameterisation for mesoscale models. *Boundary-Layer Meteorology*, 104(2), 261–304. https://doi.org/10.1023/a:1016099921195
- Miller, N. G., Gabe, J., & Sklarz, M. (2019). The impact of water front location on residential home values considering flood risks. Journal of Sustainable Real Estate, 11(1), 84–107. https://doi.org/10.22300/1949-8276.11.1.84

- Miller, S. T. K., Keim, B. D., Talbot, R. W., & Mao, H. (2003). Sea breeze: Structure, forecasting, and impacts. *Reviews of Geophysics*, 41(3). https://doi.org/10.1029/2003rg000124
- Naughton, M. P., Henderson, A., Mirabelli, M. C., Kaiser, R., Wilhelm, J. L., Kieszak, S. M., et al. (2002). Heat-related mortality during a 1999 heat wave in Chicago. American Journal of Preventive Medicine, 22(4), 221–227. https://doi.org/10.1016/s0749-3797(02)00421-x
- NOAA National Centers for Environmental Information. (2021). Climate at a glance: Global time Series. Retrieved from https://www.ncdc.noaa.gov/cag/
- Oleson, K. W., Monaghan, A., Wilhelmi, O., Barlage, M., Brunsell, N., Feddema, J., et al. (2015). Interactions between urbanization, heat stress, and climate change. *Climatic Change*, 129(3), 525–541. https://doi.org/10.1007/s10584-013-0936-8
- Ono, M., & Tonouchi, M. (2014). Estimation of wet bulb globe temperature using generally measured meteorological indices. *Japanese Journal of Biometeorology*, 50(4), 147–157.
- Paschalis, A., Chakraborty, T., Fatichi, S., Meili, N., & Manoli, G. (2021). Urban forests as main regulator of the evaporative cooling effect in cities. AGU Advances, 2(2). https://doi.org/10.1029/2020AV000303
- Pekel, J.-F., Cottam, A., Gorelick, N., & Belward, A. S. (2016). High-resolution mapping of global surface water and its long-term changes. *Nature*, 540(7633), 418–422. https://doi.org/10.1038/nature20584
- Qian, Y., Chakraborty, T., Li, J., Li, D., He, C., Sarangi, C., et al. (2022). Urbanization impact on regional climate and extreme weather: Current understanding, uncertainties, and future research directions. Advances in Atmospheric Sciences, 39(6), 819–860. https://doi.org/10.1007/ s00376-021-1371-9
- Qian, Y., Yang, Z., Feng, Z., Liu, Y., Gustafson, W. I., Berg, L. K., et al. (2020). Neglecting irrigation contributes to the simulated summertime warm-and-dry bias in the central United States. Npj Climate and Atmospheric Science, 3(1), 1–10. https://doi.org/10.1038/s41612-020-00135-w
- Raymond, C., Matthews, T., & Horton, R. M. (2020). The emergence of heat and humidity too severe for human tolerance. *Science Advances*, 6(19), eaaw1838. https://doi.org/10.1126/sciadv.aaw1838
- Romitti, Y., Sue Wing, I., Spangler, K. R., & Wellenius, G. A. (2022). Inequality in the availability of residential air conditioning across 115 US metropolitan areas. *PNAS Nexus*, 1(4), pgac210. https://doi.org/10.1093/pnasnexus/pgac210
- Roth, M., Oke, T. R., & Emery, W. J. (1989). Satellite-derived urban heat islands from three coastal cities and the utilization of such data in urban climatology. *International Journal of Remote Sensing*, 10(11), 1699–1720. https://doi.org/10.1080/01431168908904002
- Rouse, J. W., Haas, R. H., Schell, J. A., Deering, D. W., & Harlan, J. C. (1974). Monitoring the vernal advancement and retrogradation (green wave effect) of natural vegetation (p. 371). NASA/GSFC Type III Final Report.
- Salamanca, F., Krpo, A., Martilli, A., & Clappier, A. (2010). A new building energy model coupled with an urban canopy parameterization for urban climate simulations—Part I. Formulation, verification, and sensitivity analysis of the model. *Theoretical and Applied Climatology*, 99(3), 331–344. https://doi.org/10.1007/s00704-009-0142-9
- Sarangi, C., Qian, Y., Li, J., Ruby Leung, L., Chakraborty, T., & Liu, Y. (2021). Urbanization amplifies nighttime heat stress on warmer days over the US. *Geophysical Research Letters*, 48(24), e2021GL095678. https://doi.org/10.1029/2021gl095678
- Schwab, D. J., Leshkevich, G. A., & Muhr, G. C. (1992). Satellite measurements of surface water temperature in the Great Lakes: Great Lakes Coastwatch. Journal of Great Lakes Research, 18(2), 247–258. https://doi.org/10.1016/s0380-1330(92)71292-1
- Sharma, A., Hamlet, A. F., Fernando, H. J. S., Catlett, C. E., Horton, D. E., Kotamarthi, V. R., et al. (2018). The need for an integrated land-lake-atmosphere modeling system, exemplified by North America's Great Lakes region. *Earth's Future*, 6(10), 1366–1379. https://doi. org/10.1029/2018EF000870
- Sharma, A., Wuebbles, D. J., Kotamarthi, R., Calvin, K., Drewniak, B., Catlett, C. E., & Jacob, R. (2020). Urban-scale processes in high-spatial-resolution Earth system models. *Bulletin of the American Meteorological Society*, 101(9), E1555–E1561. https://doi.org/10.1175/ BAMS-D-20-0114.1
- Shepherd, J. M., & Burian, S. J. (2003). Detection of urban-induced rainfall anomalies in a major coastal city. *Earth Interactions*, 7(4), 1–17. https://doi.org/10.1175/1087-3562(2003)007<0001:douira>2.0.co;2
- Sherwood, S. C. (2018). How important is humidity in heat stress? Journal of Geophysical Research: Atmospheres, 123(21), 11808–11810. https://doi.org/10.1029/2018JD028969
- Sherwood, S. C., & Huber, M. (2010). An adaptability limit to climate change due to heat stress. Proceedings of the National Academy of Sciences of the United States of America, 107(21), 9552–9555. https://doi.org/10.1073/pnas.0913352107
- Skamarock, W. C., & Klemp, J. B. (2008). A time-split nonhydrostatic atmospheric model for weather research and forecasting applications. Journal of Computational Physics, 227(7), 3465–3485. https://doi.org/10.1016/j.jcp.2007.01.037
- Stewart, I. D., Krayenhoff, E. S., Voogt, J. A., Lachapelle, J. A., Allen, M. A., & Broadbent, A. M. (2021). Time evolution of the surface urban heat island. *Earth's Future*, 9(10). https://doi.org/10.1029/2021EF002178
- Stipanuk, G. S. (1973). Algorithms for generating a SKEW-T, log p diagram and computing selected meteorological quantities. Army Electronics Command Fort Monmouth NJ.
- Tan, H., Kotamarthi, R., Wang, J., Qian, Y., & Chakraborty, T. (2022). Impact of different roofing mitigation strategies on near-surface temperature and energy consumption over the Chicago metropolitan area during a heatwave event. *Science of the Total Environment*, 860, 160508. https://doi.org/10.1016/j.scitotenv.2022.160508
- Theeuwes, N. E., Solcerová, A., & Steeneveld, G. J. (2013). Modeling the influence of open water surfaces on the summertime temperature and thermal comfort in the city: Surface water and urban temperatures. *Journal of Geophysical Research: Atmospheres*, 118(16), 8881–8896. https://doi.org/10.1002/jgrd.50704
- Thompson, G., Rasmussen, R. M., & Manning, K. (2004). Explicit forecasts of winter precipitation using an improved bulk microphysics scheme. Part I: Description and sensitivity analysis. *Monthly Weather Review*, 132(2), 519–542. https://doi.org/10.1175/1520-0493(2004)132<0519:efowpu>2.0.co;2
- Tibbetts, J. (2002). Coastal cities: Living on the edge. Environmental Health Perspectives, 110(11), A674–A681. https://doi.org/10.1289/ ehp.110-a674
- Turner, V. K., Rogers, M. L., Zhang, Y., Middel, A., Schneider, F. A., Ocón, J. P., et al. (2022). More than surface temperature: Mitigating thermal exposure in hyper-local land system. *Journal of Land Use Science*, 17(1), 79–99. https://doi.org/10.1080/1747423X.2021.2015003
- Venter, Z. S., Chakraborty, T., & Lee, X. (2021). Crowdsourced air temperatures contrast satellite measures of the urban heat island and its mechanisms. *Science Advances*, 7(22), eabb9569. https://doi.org/10.1126/sciadv.abb9569
- Wang, J., Qian, Y., Pringle, W., Chakraborty, T., Hetland, R., Yang, Z., & Xue, P. (2023). Contrasting effects of lake breeze and urbanization on heat stress in Chicago metropolitan area. Urban Climate, 48, 101429. https://doi.org/10.1016/j.uclim.2023.101429
- Wang, J., Xue, P., Pringle, W., Yang, Z., & Qian, Y. (2022). Impacts of lake surface temperature on the summer climate over the Great Lakes region. *Journal of Geophysical Research: Atmospheres*, 127(11). https://doi.org/10.1029/2021JD036231
- Wang, W., Chakraborty, T., Xiao, W., & Lee, X. (2021). Ocean surface energy balance allows a constraint on the sensitivity of precipitation to global warming. *Nature Communications*, 12(1), 2115. https://doi.org/10.1038/s41467-021-22406-7

- Wang, Z., Song, J., Chan, P. W., & Li, Y. (2021). The urban moisture island phenomenon and its mechanisms in a high-rise high-density city. International Journal of Climatology, 41(S1), E150–E170. https://doi.org/10.1002/joc.6672
- Wang, Z.-H., Bou-Zeid, E., Au, S. K., & Smith, J. A. (2011). Analyzing the sensitivity of WRF's single-layer urban canopy model to parameter uncertainty using advanced Monte Carlo simulation. *Journal of Applied Meteorology and Climatology*, 50(9), 1795–1814. https://doi. org/10.1175/2011jamc2685.1
- Willett, K. M., & Sherwood, S. (2012). Exceedance of heat index thresholds for 15 regions under a warming climate using the wet-bulb globe temperature. *International Journal of Climatology*, 32(2), 161–177. https://doi.org/10.1002/joc.2257
- Wu, X., Zhang, L., & Zang, S. (2019). Examining seasonal effect of urban heat island in a coastal city. PLoS One, 14(6), e0217850. https://doi. org/10.1371/journal.pone.0217850
- Yang, J., Xin, J., Zhang, Y., Xiao, X., & Xia, J. C. (2022). Contributions of sea-land breeze and local climate zones to daytime and nighttime heat island intensity. *Npj Urban Sustainability*, 2(1), 1–11. https://doi.org/10.1038/s42949-022-00055-z
- Yang, Q., Xu, Y., Tong, X., Huang, X., Liu, Y., Chakraborty, T., et al. (2023). An adaptive synchronous extraction (ASE) method for estimating intensity and footprint of surface urban heat islands: A case study of 254 North American cities. *Remote Sensing of Environment*, 297, 113777. https://doi.org/10.1016/j.rse.2023.113777
- Yin, J., Schlesinger, M. E., & Stouffer, R. J. (2009). Model projections of rapid sea-level rise on the northeast coast of the United States. *Nature Geoscience*, 2(4), 262–266. https://doi.org/10.1038/ngeo462
- Zhao, L., Lee, X., Smith, R. B., & Oleson, K. (2014). Strong contributions of local background climate to urban heat islands. *Nature*, 511(7508), 216–219. https://doi.org/10.1038/nature13462
- Zhao, L., Oleson, K., Bou-Zeid, E., Krayenhoff, E. S., Bray, A., Zhu, Q., et al. (2021). Global multi-model projections of local urban climates. *Nature Climate Change*, 11(2), 152–157. https://doi.org/10.1038/s41558-020-00958-8